\begin{table}%t5
\caption{Comparison of the $V$ magnitude to the published broadband data.}\label{CompV67}
%\centering
\small
\begin{tabular}{lcr@{.}lcr}
\hline \hline
Author & & $\Delta V_{\rm our-oth}$ & $\sigma$ & $N$ \\
\hline
\cite{Eggen}  & photoel.& $-$0.02 & 0.04 & 173 \\
\cite{Sanders}& photoel.&    0.01 & 0.05 & 225 \\
\cite{Gilli}  & CCD     & $-$0.02 & 0.03 & 136 \\   
\cite{Mont}   & CCD     & $-$0.01 & 0.05 & 967 \\
\cite{Kim}    & CCD     & $-$0.01 & 0.04 & 89 \\ 
\cite{Hend}   & CCD     &    0.01 & 0.06 & 306 \\    
\cite{Sand}   & CCD     &    0.01 & 0.03 & 154 \\ 
\hline
\end{tabular}
\end{table}